Author's response to reviews

Title: Communication, perception and behaviour during a natural disaster involving a 'Do Not Drink' and a subsequent 'Boil Water' notice: a postal questionnaire study

Authors:

Gabriella Rundblad (gabriella.rundblad@kcl.ac.uk)
Olivia Knapton (olivia.knapton@kcl.ac.uk)
Paul R Hunter (paul.hunter@uea.ac.uk)

Version: 4 Date: 22 October 2010

Author's response to reviews:

Changes made:

Reference 40 - as requested, we have provided a functioning link to this website.

Line numbers - as requested, we have removed these.

Highlighting/tracking - as requested, we have removed these.

References - as requested, we have corrected references 14, 15, 17, 18, 19 and 20. We have also corrected reference 16.